Author’s response to reviews

Title: Type D personality is a predictor of prolonged acute brain dysfunction (delirium/coma) after cardiovascular surgery

Authors:

Yujiro Matsuishi (0326yuijiro@gmail.com)
Nobutake Shimojo (nkshimojo@md.tsukuba.ac.jp)
Takeshi Unoki (iwhyh1029@gmail.com)
Hideaki Sakuramoto (gongehead@yahoo.co.jp)
Chiho Tokunaga (chiho-t@md.tsukuba.ac.jp)
Yasuyo Yoshino (may2010@kantogakuin.ac.jp)
Haruhiko Hoshino (s1630445@u.tsukuba.ac.jp)
Akira Ouchi (aoouchi@times.hosp.tsukuba.ac.jp)
Satoru Kawano (kawano_s@md.tsukuba.ac.jp)
Hiroaki Sakamoto (sakamotoh@md.tsukuba.ac.jp)
Yuji Hiramatsu (yuji3@md.tsukuba.ac.jp)
Yoshiaki Inoue (yinoue@md.tsukuba.ac.jp)

Version: 2 Date: 18 Nov 2018

Author’s response to reviews:

November 18, 2018

Dear Editor,

Dr. Williams

We wish to express our strong appreciation to the Reviewers for their insightful comments on our manuscript. Their comments have helped us significantly improve the manuscript.
Here, we submit the revised version of the manuscript entitled “Type D personality is a predictor of reduced delirium/coma-free days after cardiac surgery” to be considered for possible publication in BMC Psychology. All authors have contributed sufficiently to the work and no conflict of interest exists either on findings and other financial funding.

We believe that the content of our paper will be of particular interest to the readers of BMC Psychology.

Thanks in advance for kind consideration of our paper.

Sincerely yours,

Prof. Yoshiaki Inoue
Department of Emergency and Critical Care Medicine, Faculty of Medicine, University of Tsukuba, Tsukuba, Ibaraki, 305-8575, Japan.

Phone: +81-29-853-5633.
FAX: +81-29-853-3092
E-mail: yinoue@md.tsukuba.ac.jp

REVIEWERS' COMMENTS and RESPONSES:

To Reviewer #1
I appreciate your kind response and I will greatly appreciate your further guidance.